



Article Optimized Configuration of Diesel Engine-Fuel Cell-Battery Hybrid Power Systems in a Platform Supply Vessel to Reduce CO₂ Emissions

Giovani T. T. Vieira ^{1,2,*}, Derick Furquim Pereira ¹, Seyed Iman Taheri ³, Khalid S. Khan ¹, Mauricio B. C. Salles ¹, Josep M. Guerrero ² and Bruno S. Carmo ⁴

- ¹ Laboratory of Advanced Electric Grids (LGrid), Escola Politécnica, University of São Paulo, Av. Prof. Luciano Gualberto, Travessa 3 nº 158, Butantã CEP, Sao Paulo 05508-010, SP, Brazil; dfurquim@usp.br (D.F.P.); khalid4134@gmail.com (K.S.K.); mausalles@usp.br (M.B.C.S.)
- ² Center for Research on Microgrids (CROM), Department of Energy Technology, Aalborg University, Pontoppidanstræde, 111, 9220 Aalborg, Denmark; joz@et.aau.dk
- ³ Department of Electrical and Computer Engineering, University of Central Florida, 4328 Scorpius Street, Orlando, FL 32816, USA; s.iman.taheri@knights.ucf.edu
- ⁴ Department of Mechanical Engineering, Escola Politécnica, University of São Paulo, Av. Prof. Mello Moraes, 2231, Butantã CEP, Sao Paulo 05508-030, SP, Brazil; bruno.carmo@usp.br
- * Correspondence: giovanigtvieira@usp.br; Tel.: +55-15-98163-1409

Abstract: The main objective of this paper is to select the optimal configuration of a ship's power system, considering the use of fuel cells and batteries, that would achieve the lowest CO₂ emissions also taking into consideration the number of battery cycles. The ship analyzed in this work is a Platform Supply Vessel (PSV) used to support oil and gas offshore platforms transporting goods, equipment, and personnel. The proposed scheme considers the ship's retrofitting. The ship's original main generators are maintained, and the fuel cell and batteries are installed as complementary sources. Moreover, a sensitivity analysis is pursued on the ship's demand curve. The simulations used to calculate the CO₂ emissions for each of the new hybrid configurations were developed using HOMER software. The proposed solutions are auxiliary generators, three types of batteries, and a protonexchange membrane fuel cell (PEMFC) with different sizes of hydrogen tanks. The PEMFC and batteries were sized as containerized solutions, and the sizing of the auxiliary engines was based on previous works. Each configuration consists of a combination of these solutions. The selection of the best configuration is one contribution of this paper. The new configurations are classified according to the reduction of CO2 emitted in comparison to the original system. For different demand levels, the results indicate that the configuration classification may vary. Another valuable contribution of this work is the sizing of the battery and hydrogen storage systems. They were installed in 20 ft containers, since the installation of batteries, fuel cells and hydrogen tanks in containers is widely used for ship retrofit. As a result, the most significant reduction of CO_2 emissions is 10.69%. This is achieved when the configuration includes main generators, auxiliary generators, a 3,119 kW lithium nickel manganese cobalt (LNMC) battery, a 250 kW PEMFC, and 581 kg of stored hydrogen.

Keywords: fuel cell; ship power systems; diesel engine; Li-ion battery; hybrid power systems; hydrogen storage

1. Introduction

In recent years, greenhouse gas (GHG) emissions from the shipping sector have concerned the international community. In 2018, the United Nations' International Maritime Organization (IMO) agreed to reduce their GHG emissions of global shipping by 50% before 2050 compared to their levels in 2008 [1].

Even the Paris Agreement and the Kyoto Protocol, which did not directly set any objectives to reduce emissions, have brought up shipping emissions in discussions [2].



Citation: Vieira, G.T.T.; Pereira, D.F.; Taheri, S.I.; Khan, K.S.; Salles, M.B.C.; Guerrero, J.M.; Carmo, B.S. Optimized Configuration of Diesel Engine-Fuel Cell-Battery Hybrid Power Systems in a Platform Supply Vessel to Reduce CO₂ Emissions. *Energies* **2022**, *15*, 2184. https:// doi.org/10.3390/en15062184

Academic Editor: Il-Yop Chung

Received: 18 February 2022 Accepted: 11 March 2022 Published: 17 March 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Figure 1 shows a comparison between the CO_2 emissions for the years of 2018 [3] and 2015 [4]. Considering that Germany is part of the European Union, this ranking includes the fifteen largest emitters in the world, where the shipping sector stands in the eighth place. Furthermore, this sector also had the seventh-largest increase (more than 6%) from 2015 to 2018. Regarding the fact that the emissions from countries are covered and analyzed by the agreements mentioned, this work aims at analyzing possible solutions to reduce the CO_2 emissions of ships.



Figure 1. Global CO₂ emissions ranking comparing the emissions data from 2018 [3] with the data from 2015 [4].

Figure 1 shows a comparison between the CO_2 emissions for the years of 2018 [3] and 2015 [4], of the fifteen largest emitters in the world. Since Germany's emission are already included in those of the European Union, the shipping sector alone is the eighth largest CO_2 emitter in the world. Furthermore, the sector also had the seventh largest increase (more than 6%) from 2015 to 2018. Regarding the fact that the emissions from countries are covered and analyzed by the agreements mentioned, this work aims at analyzing possible solutions to reduce the CO_2 emissions of ships.

The emission data presented in annual reports [3,4] show that there is still a great effort to be made regarding the reduction of GHG in the shipping sector. In this sense, the use of different hybrid technologies in ship power systems is essential to guarantee a more ecological future. There are many feasible solutions available that could be employed in shipboard power systems and potentially reduce GHG emissions: biofuels [5], batteries [6,7], fuel cells [8], and direct current (dc) distribution [9], among others. Considering that the IMO aims at reducing the emissions related to ships, and these emissions may have a limit that when reached can implicate in taxes, the use of hybrid power systems is attractive to shipbuilders and ship companies to increase their profit by reducing their costs related to fuel and possibly avoiding paying taxes due to an excess of emissions.

Ship emission estimation methods are described in [10–14]. These methods may have drawbacks based on how the emissions are measured. In [12], it is mentioned that the more realistic approach would be to estimate emissions based on a ship's actual speed. For ships with only main engines, the relationship between ship speed and power output of the generators is simple [15]. When auxiliary generators, fuel cells, and batteries are

considered, this relationship is much more complex. More detailed models, tools, and studies are needed to be able to more accurately measure the effects that ship power systems' hybridization will have on the GHG emission levels. Currently, there are very few simulation and design optimization tools available for hybrid power systems in the shipping sector [16]. General applications frequently use HOMER software for the optimal design of hybrid power grids [17–20], but HOMER can also be used for maritime applications [7,21–25]. As seen in previous works [7,23], HOMER can be used to estimate the CO_2 emissions of ships. Moreover, the main and auxiliary engines' load factors will be set according to the software's optimal energy-dispatch algorithm.

As the main objective, this paper selects an optimal configuration of a ship's power system, considering the use of fuel cells and batteries, that would achieve the lowest CO_2 emissions, also taking into consideration the number of battery cycles. HOMER PRO is used to calculate the emissions and the battery cycles of various configurations and three different levels of demand. Then, the best configuration that can be composed of fuel cells, batteries, and main and auxiliary generators is found based on these criteria. The ship analyzed in this work is a platform supply vessel (PSV). One main concern is that the solutions applied to ship power systems usually require a new ship to achieve the reductions estimated. This requires a higher amount of money and may take a long time to present benefits. For this reason, the solutions analyzed in this paper are designed for a ship's retrofit. Therefore, this method can be used to assess if the CO_2 reductions achieved can pay for the cost of installation and the loss of space required for the new equipment. Another important contribution is the sizing of the battery and hydrogen storage systems. This work sizes them for an installation in 20 ft containers commonly used in PSVs. Moreover, the proposed method performs a sensitivity analysis considering the variation of the ship's demand curve. We considered that the PSV has a fixed scheduled routine, which means that, independent of the weather conditions, the ship must arrive on time. The variation in weather and sea condition interferes with the power demand required. When there is a rough sea with bad weather conditions, the demand will be higher. In a calm sea with good weather conditions, the same ship would require a lower demand to arrive on time.

The remaining sections of this paper are organized as follows. Section 2 describes the power system dispatch strategy and CO_2 emissions calculation in the HOMER software. Section 3 presents the main characteristics of the PSV under study, as well as the demand curve during a typical mission. Section 4 presents the alternative power generation components considered in this work. In Section 5, data on the fuel cell, diesel generator and batteries are presented together with hydrogen-storage-sizing and battery-system-sizing information. Section 6 presents the results of the CO_2 emissions estimations. Finally, Section 7 summarizes the main contributions of this work and presents the conclusions.

2. Dispatch Strategy and CO₂ Emission Calculation

The simulations were performed with HOMER, a commercial software with a proprietary derivative-free optimization method that provides an optimal dispatch of the energy available to supply the demand. The objective function of HOMER's optimal dispatch is the system's net present cost, i.e., the sum of the costs of installing and operating all the system's components over the project's lifetime. As HOMER's optimization is used to determine the impact that each configuration will have on the emissions, most of the costs that would have the highest impact on emissions were set to zero. Battery costs and hydrogen costs were set to zero to force HOMER to use them as much as possible. The aims to use these sources with no emissions as long as possible, in order to ensure as much emission reduction as possible, no matter the cost. Furthermore, the operations and maintenance (O&M) cost of the converters was also set to zero, and HOMER's default diesel fuel price of 1\$/L was maintained.

Beside the costs, HOMER has other inputs, such as fuel consumption curves for the generators, battery round-trip efficiency, converter efficiency, battery initial state of charge

(SoC), minimum SoC, and the set-point SoC that is the maximum SoC allowed during the simulation. The choices of the minimum and maximum SoC values were made according to the impact that the depth of discharge (DoD) can cause in a battery's life. The DoD is the difference between the maximum and the minimum SoC that the battery can reach during charge and discharge, respectively.

In [26], there is a study investigating which would be the optimum DoD for a 24 kWh battery. The optimization method determined a DoD of 70.1%, considering a premise that the system would have 100% availability. In [27], the DoD that would provide the lowest total cost for a battery installed in a round-trip electric bus route was determined through convex programming (CP).

When battery degradation is considered, the DoD estimated is 53.1%. Without the degradation, the DoD moves to 46.6%.

The optimal DoD values obtained in [26,27] were, respectively, 70.1% and 46.6%. Based on this variation, considering cost reduction and availability, the DoD chosen in this paper is 60%, and the maximum and minimum SoCs are set to 80% and 20%, respectively. The present study considers a new and fully charged battery at the beginning of the simulation. Therefore, from the simulation start until the moment that the battery SoC is under 80%, no charge is allowed. Moreover, the battery is not allowed to discharge more if the SoC reaches 20%. The battery dispatch is chosen according to the cycle-charging dispatch of HOMER, in which a set-point SoC is applied to the cycle-charging strategy. The value applied for this set-point is 80%, which is the same used for the maximum SoC. In this way, the SoC will always be below the set-point during the simulation, and, if in a previous time step, the battery was not feeding the primary load, in the current time step HOMER will avoid discharging the storage. During the charging cycle, a generator needs to produce enough energy to serve the primary load and charge the storage bank. When a charging cycle starts, the battery is charged until it reaches the set point state of charge. In short, when the battery starts to discharge, it does so until its SoC reaches 20%. When it starts to recharge, it does so until the SoC reaches 80%.

Since HOMER optimizes the net present cost over its lifetime, it requires yearly demands to perform the simulations. In this work, the demand lasts for only 113 h, which is the ship's mission duration, and, thus, the yearly demand curve is filled with zeros after the 113th hour. Since HOMER requires a load to dispatch a controllable power source, generally, there is no emission in the first hour with a demand level equal to zero. An exception is when the battery is charging in the previous time-step; then, the next hour after the end of the demand can be used to assess the battery until the set-point.

The calculation of CO_2 emissions by HOMER is divided into three steps that are run internally by the software. Firstly, before the simulation, HOMER determines the emission factor of CO_2 , representing the amount of CO_2 emitted in kilograms per unit of fuel consumed in liters. The user directly specifies the carbon monoxide and unburned hydrocarbons emissions factors. Then, HOMER assumes that all the carbon in the fuel that was not emitted as carbon monoxide or unburned hydrocarbons will be emitted as carbon dioxide. The default emissions factors of diesel fuel given by the HOMER software and considered in this work are shown in Table 1. In the second step, HOMER performs the simulation and evaluates the fuel consumed. After that, in the third step, HOMER multiplies the fuel consumed by the emission factor to obtain the amount of pollutant emitted.

Table 1. Emissions Factors.

Pollutant	Unit	Value
Carbon monoxide	(g/L)	6.5
Unburned hydrocarbons	(g/L)	0.72

Power Demand (MW)

3. General Characteristics

This section describes how the demand was designed and highlights the importance of pursuing a sensitivity analysis on the demand curve. Moreover, this section also describes the main components of the PSV power system in the base case and how the solutions analyzed in this work can be connected to the system.

3.1. Demand Design

The demand curve used in this paper is presented in Figure 2, highlighted as a continuous red line and represents a standard PSV routine based on [28]. This normal demand profile has been used in many works [7,29,30]. The variation of +10%, depicted in dashed black line, and the variation of -10%, presented in dotted blue line, are performed to pursue the sensitivity analysis. Besides that, the three levels of demand analyzed in this paper have five major parts, which are divided by vertical lines in Figure 2: loading in port, laden voyage, dynamically positioned (DP) operation, partial load voyage, and stand-by.



Figure 2. Standard PSV power demand considering different parts of the mission pursued (red curve) and variations considered for the sensitivity analysis (+10% original power demand—dashed black line, -10% original power demand—dotted blue line).

The average demands of each part of the PSV mission described hereafter refer to the red line in Figure 2. The vessel starts the trip being loaded at the port with an average demand of 482 kW. Once it is fully loaded, it starts the voyage towards the platform with an average demand of 6100 kW. Then, the ship stops by the platform to move its loads to the platform and receive some loads from the platform; this operation is named DP. It is the most dangerous operation that a ship performs, and the average demand power

required during DP operation is around 1250 kW. The voyage back to the port with the ship partially loaded is named the partial load voyage, and its average demand is 5040 kW. The last part, called standby operation, describes the demand when the vessel needs to wait for the port to be ready. The standby operation has an average power of 335 kW and could also represent when the ship waits to begin a DP operation.

In this paper, a sensitivity analysis is pursued, varying the demand curve by $\pm 10\%$. It aims to represent the impact that variations on speed, cargo loaded, wind direction, and wave force may cause on the demand curve. This sensitivity analysis will help investigate the amount of CO₂ emission reduction that each configuration can obtain, considering different demand levels. The demand curve varies due to sea and weather conditions. Considering that the ship's mission has scheduled hours to arrive on each platform, moving the ship from one point to another in a rough sea with bad weather conditions would require a higher demand. On the other hand, a calm sea with good weather conditions allows a ship to travel with a lower demand.

The strength and direction of the wind, the force of the waves, and how the operator drives the ship are three important factors that will determine the power demand. The sensitivity curve shown in Figure 2 aims at covering possible variations that the demand curve may have. As shown in Figure 2, the demand curve highlighted in red is considered the normal demand. This curve was increased by 10% and reduced by 10% to produce the variations.

3.2. PSV Power System

The alternative solutions used in this paper to reduce CO_2 emissions consist of fuel cells, batteries, and auxiliary diesel engines, as presented in the diagram of the PSV power system in Figure 3. The base case is displayed in black and consists of four C280-6 Caterpillar 1820 kW diesel generators [31], one 750 kW base load, one 300 kW service load, two 900 kW bow thrusters, and two 2 MW azimuth thrusters. The alternative solutions to reduce CO_2 emissions are highlighted in different colors in Figure 3. The two auxiliary generators are shown in red and consist of two C18 Caterpillar 480 kW [32]. The advanced battery system is presented in green. It can represent any of the batteries analyzed in this work: lithium titanate-oxide (LTO), lithium iron phosphate (LFP), or lithium nickel manganese cobalt oxide (LNMC). Finally, Figure 3 depicts in blue a 250 kW proton exchange membrane fuel cell (PEMFC), which is fed with pure hydrogen.

The alternative solutions to reduce CO_2 emissions used in this work consist of fuel cells, batteries, and auxiliary diesel engines. The power system from [33] is used as the base case and highlighted in black in Figure 3. The basic configuration consists of four C280-6 Caterpillar 1820 kW diesel generators [31], one 750 kW base load, one 300 kW service load, two 900 kW bow thrusters, and two 2MW azimuth thrusters. The alternative solutions to reduce CO_2 emissions are highlighted in different colors in Figure 3. The two auxiliary generators are shown in red and consist of two C18 Caterpillar 480 kW [32]. The advanced battery system is presented in green. It can represent any of the batteries analyzed in this work: lithium titanate-oxide (LTO), lithium iron phosphate (LFP), or lithium nickel manganese cobalt oxide (LNMC). Finally, Figure 3 depicts in blue a 250 kW proton exchange membrane fuel cell (PEMFC), which is fed with pure hydrogen.

This study was performed under the assumption that the power system in Figure 3 has control systems that are installed to allow the operation in DP with the bus-tie breaker closed. In this way, the engines can operate at a more efficient level. Without such control systems, the operation with the bus closed may lead to a major problem in the case of a failure or a fault [34].

Based on Figure 3, the set of configurations used in this work can be with or without the PEM fuel cells, with or without one of the batteries (LNMC, LTO, and LFP), and with or without auxiliary generators. All the power system's configurations are compared with the base case that includes only the four 1.82 MW main generators. The analysis neither considers the batteries' aging nor the energy required by the PEMFC cooling system.



Figure 3. Schematic three-phase diagram of the PSV power system.

4. Technical Solutions

4.1. Auxiliary Diesel Generators

One of the main reasons for using auxiliary generators on ships is their level of technological maturity. The use of auxiliary generators can be seen as the more mature solution, since they are also operated by diesel, and there is a strong background in diesel engine maintenance in the shipping sector. Furthermore, as seen in Figure 2, the PSV demand curve has a considerable variation in its power level. Since the diesel engines' highest efficiency is usually achieved at 80% of their power output, main diesel generators are not the best alternative to power the ship during a low load operation. Therefore, the use of auxiliary generators with reduced rated power (480 kW) is more efficient during low loads, leading to a reduction in fuel consumption and, consequently, in CO₂ emissions. Auxiliary generators will typically operate with more than 40% of their capacity even when the ship is in loading in port and on standby, reducing ignition problems and poor combustion [35].

4.2. Batteries

This paper analyzes the impact of three types of lithium-ion batteries: (a) lithium titanate-oxide (LTO), (b) lithium iron phosphate battery (LFP), and (c) lithium nickel manganese cobalt oxide (LNMC). A comparison of these types of batteries is portrayed in Figure 4. Although LTO batteries present the best numbers for the number of cycles and round-trip efficiency, they have the smallest specific energy and consequently weigh more to have the same energy as the LFP and the LNMC batteries.



Figure 4. Comparison of LFP, LTO, and LNMC considering Maximum and Minimum Temperature of Operation [36], Energy Installation Cost [37], Specific Energy [38], Number of Cycles [36], and Round-trip Efficiency [37]. The number of cycles is considered to be the number of charges and discharges that will reduce the battery capacity to 80% of its rated capacity. The values for Minimum and Maximum Temperatures, Round-trip Efficiency, and Number of Cycles do not start from 0 because of graphical limitations.

The fast response and the possibility of operating the battery as a load or as a generator according to its charge and discharge cycle, respectively, are the main points increasing the interest of shipbuilders and ship operators in energy storage systems. The fast response can help to quickly re-energize the ship after a blackout, and optimizing the batteries' charge and discharge cycles may result in more efficient use of the generators. Moreover, in low load operation cases, the diesel generators may be disconnected, and the ship power systems can be fed by batteries for some hours or minutes, depending on their storage capacity and SoC. Batteries are usually kept in 20 ft containers with converters and other thermal and dispatch management equipment.

Battery costs are still falling, due to the increase of the battery market and the use of high energy density cathodes. As such, a constraint that should be discussed is their life expectancy. Operating the battery with a lower depth of discharge DoD is an excellent alternative to increase its life expectancy and reduce its capacity loss [39,40]. Another way of analyzing the relationship between cost and life expectancy is through the Levelized Cost Of Energy (LCOE), which measures the average net present worth of the energy generated over its lifetime. According to [41], the LCOE for lithium-ion batteries has fallen by 35% since the first half of 2018, reaching \$187 per megawatt-hour. In terms of battery price, considering a battery cell or a pack, the price has fallen 87% reaching \$156/kWh in 2019.

One important parameter is the battery's state of health (SoH), which reflects its general condition, and is given by:

$$SoH = 100 \frac{C_{Ncycles}}{C_{new \ battery}},\tag{1}$$

where $C_{Ncycles}$ is the available capacity after *Ncycles* battery cycles and $C_{new \ battery}$ is the nominal capacity. A new battery has a SoH of 100%, which begins to fade once the battery starts cycling. According to [43], when the SoH decreases to 80%, the battery is no longer indicated for vehicle application. This number of cycles after which the battery reaches the end of its useful life depends on various factors, such as temperature, DoD, and C-rate. As shown in Figure 4, the cycles for LNMC, LFP, and LTO are 3000, 3400, and 15,000, respectively. In [36], it is shown that these cycles are determined for a DoD of 100% and a temperature of 25 °C; the c-rate varies for each battery.

4.3. Fuel Cells and Hydrogen Storage

Unlike batteries, which are rechargeable and independent energy storage devices, fuel cells (FCs) only perform energy conversion and require a continuous fuel supply to generate electricity. The energy in this case is stored in the form of the fuel itself. Therefore, when considering the use of FCs in hybrid maritime power systems, the fuel type and also the fuel storage solution must also be taken into account.

Among the different types of FCs, the proton-exchange membrane fuel cell (PEMFC) is one of the most suitable for maritime transport applications [44]. This is the type of FC with the greatest variety of applications, and it is trendy in vehicular applications [45], thanks to its solid electrolyte and operation at low temperatures (below 100 °C) [46]. The PEMFC can be fueled with pure hydrogen gas, which can be stored in high pressure [47] or metal hydride tanks [48,49], or by a natural gas reformer [50]. If natural gas is used as fuel, there would be CO and CO₂ emissions, while pure hydrogen allows an emission-free operation. This work considers using a PEMFC supplied with pure hydrogen, so the proposed design allows an emission-free operation of the FC system.

The most popular method of hydrogen storage in the industry is in high-pressure cylinders. Typically, steel cylinders are used to store and transport compressed gas at ambient temperature. It is a simple method and does not require any power to consume the stored hydrogen gas. However, it has disadvantages, such as the power consumed during the gas compression process and, mainly, safety-related concerns, since there is a risk of explosion due to high-pressures [51].

More recently, the chemical storage of hydrogen in the form of metal hydrides has become a promising solution [49]. In this method, the storage is based on hydrogen's reversible reaction with different metals, alloys, and other metallic compounds. The main advantages of metal hydrides over compressed hydrogen are their energy density (kWh/m³) and safety. The energy density of metal hydride tanks is higher than that of pressurized hydrogen and comparable to that of liquid hydrogen [52]. However, liquid storage has the disadvantage of energy costs to liquefy hydrogen [51]. In terms of safety, in metal hydride tanks, hydrogen is not stored at very high pressures. Therefore, there is no risk of an abrupt and dangerous discharge, as in the case of pressurized hydrogen.

Despite having a higher energy density (kWh/m³), metal hydride tanks have a specific energy (kWh/kg) lower than that of pressurized hydrogen tanks [52]. Thus, they are suitable for applications where there are usually not so many weight restrictions, but there are significant space restrictions, as in ships. For this reason, in addition to the high level of safety, in this paper, hydrogen storage using metal hydrides is compared with hydrogen stored in high-pressure cylinders.

5. Components Configuration

Containerized solutions are commercially available for fuel cells [53,54], batteries, and hydrogen. Therefore, each of the technical solutions evaluated in this paper was sized to be installed in 20 ft containers. The battery sizes analyzed in this paper are related to commercial cell size, as is further explained in Section 5.4. The sizing of the hydrogen container is detailed in Section 5.2.

The fuel cell was sized to 250 kW. In [55]; two 250 kW fuel cells were analyzed to reduce the pollutant emissions of an offshore platform supply vessel. In this paper only one 250 kW fuel cell is considered, because batteries and auxiliary generators are also evaluated. Reducing the number of fuel cells impacts on the number of containers installed.

Auxiliary generators were sized to 480 kW because of the demand curve and the size of the main generators. Considering the standard PSV configuration with only main diesel generators, the ship operates in low-load (\leq 40%) during the whole loading in port and standby operations. With the use of auxiliary engines, the ship operates in low-load only 28.6% of the time during loading in port and 12.5% during standby for the worst case scenario in which loads above the rated power of one auxiliary engine would not be divided equally between two auxiliary generators. Beside that, the use of two auxiliary engines can cover the maximum demands during loading in port and standby, operation with only the main engines would also cover the demand during these two parts, but the engines operate in low-load as mentioned before.

5.1. Fuel Cell Efficiency

The efficiency of a FC is calculated as the ratio between the electrical energy generated and the chemical energy supplied to the FC, which is equivalent to the heat that would be produced if the hydrogen were burned [52]. Figure 5 displays an efficiency curve that has a profile based on data from NREL's Advanced Vehicle Simulator [56,57], which states that the maximum efficiency is achieved in around 40% of the rated power. Additionally, the maximum efficiency value was obtained in [45], which states that the maximum efficiency to 60%. The PEMFC connects via an inverter to an ac bus of the ship. This study considers that the inverter's efficiency is constant and equal to 98%.



Figure 5. Efficiency curve of the PEMFC used in this work. The efficiency curve is drawn based on the profile found in [56] and on the values found in [45].

5.2. Hydrogen Storage Sizing

This work considers two types of hydrogen storage technology: metal hydrides and compressed hydrogen in high-pressure tanks. Table 2 presents the technical characteristics

of the different hydrogen storage tanks analyzed in this study. Tank A consists of a lanthanum-nickel hydride (LaNi₅H₆) tank [58], which is an AB₅-type intermetallic alloy. These alloys are one of the most popular hydride materials in hydrogen storage applications, since their hydrogen sorption properties can be adjusted through small variations in their composition [49]. In this way, it is possible to align the material's operating temperature and pressure depending on the application. Moreover, despite their lower gravimetric hydrogen storage density when compared to other metal hydrides such as the magnesium hydride (MgH₂), AB₅ alloys can operate at low temperatures (below 30 °C) [49,58]. Magnesium hydride requires higher temperatures (around 300 °C) to decompose and release hydrogen, which typically restricts its application to systems that can be thermally integrated with a high-temperature heat source, such as a SOFC [59].

Table 2. Technical characteristics of the different types of hydrogen storage tanks considered.

Tank	Α	В	С		
Technology	Metal hydride	Compressed hydrogen	Compressed hydrogen		
Туре	LaNi ₅ H ₆	IV (700 bar)	IV (500 bar)		
Hydrogen capacity (kg)	0.63	1.78	11		
Diameter (mm)	169	308	531		
Length (mm)	1460	827	2424		
Energy density (kWh/m ³)	641	962	683		

In addition to hydrogen storage in metal hydrides, this study examines two options for compressed hydrogen storage pressure: 700 bar [47] and 500 bar [60], presented in Table 2 as tanks B and C, respectively. Both B and C tanks consist of composite cylinders with a polymer liner, known as a Type IV tank. Such tanks have an operating pressure in the range of 350 to 700 bar. Therefore, such a container can transport more gas. Tank C, in Table 2, is a tank with optimized dimensions for transporting hydrogen in containers [60,61]. Despite its operating pressure of 500 bar, it stores more hydrogen than a 700-bar-capacity container, due to its optimized dimensions. However, considering the energy density, tank B outperforms tank C, which has an energy density closer to that of tank A.

The analysis carried out in this work considered the possibilities of accommodating the hydrogen tanks in a standard 20 ft container (5.90 m \times 2.35 m \times 2.39 m). Assuming the tanks can be positioned side by side in the vertical position inside the containers according to the representation shown in Figure 6, the total number of tanks that can be accommodated in each container will be given by:

$$T = \lfloor \frac{W}{d} \rfloor \lfloor \frac{L}{d} \rfloor, \tag{2}$$

where *T* is the number of tanks; *W* and *L* are, respectively, the width and length of the container; *d* is the hydrogen tank diameter; and $\lfloor \cdot \rfloor$ denotes the floor function, which takes as input a real number and outputs its integer part (greatest integer less than or equal to that real number).



Figure 6. Representation of the arrangement of the hydrogen cylinders inside a container.

The number of tanks from Table 2 that can be accommodated in a 20 ft container, and the total hydrogen storage capacity, is listed in Table 3. Considering tank A, the number of tanks stored in a container was calculated directly by (2). In the case of tank B, the number of tanks per container considered was twice that given by (2), since the length of the tank is only 827 mm. Therefore, it was considered that the container can accommodate twice the number of tanks if they are stacked. In the case of tank C, the number of tanks that can be accommodated in a 20 ft container was provided by the manufacturer in [61].

Table 3. Hydrogen storage capacity in a 20 ft container.

Tank	Nb of Tanks	Hydrogen Stored (kg)
А	442	278
В	266	486
С	52	565

Table 3 demonstrates that, using metal hydrides, a standard 20 ft container can store up to 278 kg. Despite tank B's higher energy density, a 20 ft container can store less hydrogen than tank C, mainly due to the tanks' dimensions. In the studies carried out with HOMER presented in this work, the three amounts of stored hydrogen listed in Table 3 are considered. It is important to remark that this study considers that the tanks are full at the beginning of the mission, and they are not refilled during the mission.

5.3. Diesel Generators

Figure 7 shows the efficiency curves of the main and auxiliary generators. The main generators have a rated power of 1.82 MW and consist of C280-6 diesel generators manufactured by Caterpillar [31]. The auxiliary generators have a rated capacity of 480 kW and consist of C18 diesel generators [32] from the same manufacturer. As shown in Figure 7, both have higher efficiency when they operate closer to their rated power. The data of the efficiency curves of both generators were obtained from [31,32].

In the base case, only the four 1.82 MW generators are considered. In this configuration, they must operate with a minimum power output of 10% each, i.e., none of the generators will be connected unless there is a minimum load of 182 kW. Therefore, the operation can be guaranteed when the load is lower (loading in port and standby). The use of auxiliary diesel engines reduces the emissions mainly during the low-load operation as described in [7]. With auxiliary diesel engines, the minimum power output of all diesel engines can be increased to 40%. This power corresponds to 728 kW in the main generators and to 192 kW in auxiliary ones, which is in accordance with the minimal demands required during loading in port and standby. When fuel cells and batteries are inserted in the ship power system, the minimum level of diesel engines can be set as 50%. In this configuration, batteries and fuel cells can cover loads that are under 225 kW and cannot be covered by auxiliary diesel engines.





Figure 7. Efficiency curve of the main and auxiliary diesel engines, these curves were obtained from [31,32], respectively.

5.4. Battery Sizing

The batteries were sized as described in [7]. Firstly, the volumes occupied by the cells in a pack and by packs in a container were determined to be 57% and 34%, respectively. Then, commercial cells were considered. Although all the cells evaluated in this work once existed, due to the cell development, some cells were not available commercially at the time that this paper was written, so they have no citation.

The volume of each pack was evaluated based on cell dimensions and their electrical parameters, and considering the system voltage fixed at 1000 V. Since the container volume occupied per pack was 34%, a simple division provides the number of packs in each container. Finally, power and energy ratings are given considering that the battery is operating with a C-rate of 1. When the simulations were performed, the C-rates respected the maximum limits indicated by the "recharge rates" column in Table 4. These values were taken from the cells' datasheet.

Table 5 shows the volume in m³ that one battery pack would occupy according to each cell's volume. Table 4 shows the battery parameters used to calculate the energy and the power of the batteries used in this work.

It was considered that filters, controllers, converters, energy management systems, and the equipment used to control the temperature of the container would occupy the remaining 66%.

The round trip efficiency (RTE) was kept as the values of the table. The efficiency of the battery inverter was considered constant and equal to 98%.

Parameter	RTE *	Cycle Life	Recharge Rates	Container Volume (m ³)	% of the Container Occupied with Batteries	Effective Volume (m ³)	Total Packs Container	Power Pack (W)/Energy Pack (Wh)	Power Container (kW)	Energy Container (kWh)
LTO 948 kW	96	10,000	3C-1C	33.19	34	11.28	862.147	1100	948.362	948.362
LTO 1121 kW	96	10,000	3C-1C	33.19	34	11.28	862.147	1300	1120.79	1120.79
LFP 1510 kW	92	2500	2C-1C	33.19	34	11.28	1208.01	1250	1510.01	1510.01
LFP 1284 kW	92	2500	2C-1C	33.19	34	11.28	1605.68	800	1284.5	1284.5
LNMC 2410 kW	95	2000	3C-1C	33.19	34	11.28	964.159	2500	2410.4	2410.4
LNMC 3119 kW	95	2000	3C-1C	33.19	34	11.28	623.8	5000	3119	3119

 Table 4. Container of battery parameters.

* Round trip efficiency.

Table 5.	Battery	pack	parameters.
----------	---------	------	-------------

Parameter	Diameter (mm)	Height (mm)	Cell Volume (m ³)	Nominal Voltage (V)	Cell Capacity (Ah)	Cell Energy (Wh)	System Voltage (V)	Number of Cells in Series	% of Cells in a Pack	Pack Volume (m ³)
LTO 948 kW	18.7	65.5	$1.79 imes10^{-7}$	2.4	1.1	2.64	1000	416	57	0.0131
LTO 1121 kW [62]	18.7	65.3	$1.79 imes 10^{-7}$	2.4	1.3	3.12	1000	416	57	0.0131
LFP 1510 kW	18.2	65.6	$1.71 imes 10^{-5}$	3.2	1.25	4	1000	312	57	0.0093
LFP 1284 kW [63]	18.1	49.9	$1.28 imes 10^{-5}$	3.2	0.8	2.56	1000	312	57	0.0070
LNMC 2410 kW [64]	22	65	$2.47 imes 10^{-5}$	3.7	2.5	9.25	1000	270	57	0.0117
LNMC 3119 kW [65]	26.9	65.5	$3.72 imes 10^{-5}$	3.6	5	18	1000	277	57	0.0181

6. Results

In the following subsections, the results of the configurations analyzed are shown according to the demand level. For the following analysis, the minimum loads for main generators vary according to each configuration. For configurations including only main generators, main and auxiliary generators, main generators and PEMFC, main and auxiliary generators, and PEMFC, the minimum load set is 10%. For configurations including the battery, the minimum load set is 40%.

In figures that shows the emission results, as Figure 8, the bar colour represent the amount of hydrogen stored in the tanks, and the illustrated rectangles represent the batteries. The figures are divided in four parts, each part represents the configuration results discussed in them, the configurations are named above the group of bars. The first box on the left shows the base case that includes only main generators. In this first box, the reduction obtained by the connection of fuel cells is also shown. The numbers, inside the bar, represent the percentage reduction that is achieved by the use of fuel cells when compared to the case written above the bars. The numbers above the bars represent the percentage reduction obtained by the configuration that includes the equipment names written above the bars and the PEMFC in comparison to the base case. In the first box, the numbers inside and above the bars are the same.

The second box from the left to the right shows the results of the configuration including the main generators and auxiliary generators. The reduction obtained by the comparison between this configuration and the base case is shown inside the larger rectangle. The bar above the rectangle in this second box shows two pieces of information. Firstly, inside the bar, we can see the reduction obtained from the use of fuel cells in comparison to the case including main and auxiliary generators. Secondly, above the bar, the total percentage reduction obtained by the configuration including main and auxiliary generators and PEMFC in contrast to the base case including only main generators is shown.

The same discussion did for the second box occurs for the third and the fourth boxes. For the third box, the larger rectangle represents the reduction obtained for the configuration, including the main generators and batteries. In the fourth box, the larger rectangle shows the reduction obtained by the configuration, including the main and auxiliary generators and batteries in relation to the base case.

6.1. 10% Lower Demand

Figure 8 shows the emission percentage reduced for all configurations related to the base case that has only four 1820 kW diesel generators. This percentage is shown in the vertical axis, and lines are traced through the figure to help readers to evaluate the emission reduction.

As can be seen, the most representative emission reduction is achieved for the configuration including main and auxiliary engines, a fuel cell with a hydrogen tank of 565 kg, and a LNMC-3119 kW battery. This configuration achieved an emission reduction of 10.69%, when the 10% lower demand is analyzed.

When configurations without auxiliary engines are analyzed, the configuration including main generators, a fuel cell with 565 kg of hydrogen, and the LNMC-3119 kW battery achieves the highest emission reduction. This configuration achieved a reduction of 9.98%. If the hydrogen storage is reduced to 486 kg, this same configuration also achieves the highest reduction, around 9.19%. For a hydrogen storage of 278 kg, this configuration also achieves the highest emission reduction, around 7.33%.



Figure 8. Total emission reduction results obtained from the configurations operating at 10% lower demand.

The results in Figure 8 demonstrate that, for a demand equivalent to the 10% lower demand, the emission reduction achieved by configurations including main and auxiliary generators and batteries simultaneously is not much higher than the emission reduction achieved by configurations including main and auxiliary generators or a main generator and batteries.

The same analysis can be performed in configurations including PEMFC; the emission reduction obtained in configurations including main generators, PEMFC, auxiliary generators, and batteries is close to the one obtained by configurations including main generators, PEMFC, and auxiliary generators or configurations with main generators, PEMFC, and batteries. Therefore, if only the emission reduction perspective is analyzed, it may not be worth the investment in both auxiliary generators and batteries.

A cost analysis considering the relationship between the amount invested and the reduced CO_2 ($\frac{1}{CO_2}$) emissions must be carried out later for a definitive choice between batteries or auxiliary generators. Despite the small influence of auxiliary generators in reducing total emissions, their use allows a reduction in the number of battery cycles. This discussion will be presented later.

Figure 8 also demonstrates the impact that the connection of fuel cells can cause in emission reduction. The reduction achieved by connecting a fuel cell, for the same hydrogen amount, is very close, regardless of the configuration. If we compare the emission reduction achieved by the connection of the three solutions (auxiliary engines, battery, and PEMFC) individually, we can see that batteries and auxiliary generators present a higher reduction than the use of the PEMFC with 278 kg. Hydrogen tanks with 486 kg and 565 kg achieve a higher emission reduction than batteries and auxiliary generators when connected to the base case. Additionally, a comparison between the emission reduction achieved using the hydrogen tank with 486 kg and the one obtained using 565 kg of hydrogen shows that the difference is not representative. Therefore, depending on the cost involved in storing more hydrogen, the use of 486 kg could have a higher cost–benefit relationship.

Figure 8 also shows the amount of CO₂ emitted for configurations without PEMFC. As can be seen, the use of batteries can provide a small emission reduction when connected in configurations including main and auxiliary engines. The best configuration without PEMFC is the configuration including main and auxiliary generators and the LNMC-3119 kW battery. Moreover, if configurations without PEMFC and auxiliary engines are evaluated, the highest emission reduction is achieved by the configuration including main generators and a LNMC-3119 kW battery. It is interesting to note that the configuration including only main engines and batteries presents a higher emission reduction than the configurations including only main and auxiliary engines and the configurations including only a main generator and PEMFC with 278 kg.

Finally, reductions of configurations with PEMFC and without a battery can also be measured in Figure 8; for configurations including only main generators these reductions goes from 2.66% to 5.06%. Moreover, for configurations including main and auxiliary generators, these reductions change from 6.51% to 9.09%.

Figure 9 shows the number of cycles that these batteries have in each configuration when the 10% lower demand is considered. This figure also shows that there is a connection between the number of cycles and the battery capacity. Moreover, the use of auxiliary engines provides a considerable reduction on battery cycles.

LFP batteries achieved the minimal number of cycles with the configuration, including main and auxiliary generators and without the PEMFC. The LFP-1284 kW achieved 12 cycles, and the LFP-1510 kW achieved 11 cycles.

In terms of cycles, for all configurations, the best and worst batteries were LNMC-3119 kW and LTO-948 kW, respectively. Configurations with auxiliary engines also presented a smaller difference, in terms of cycles, between the best and the worst battery. For the configuration including only main generators and batteries, this difference was 24 cycles. For the configuration including main and auxiliary generators and batteries, this difference was reduced to 14 cycles. The use of auxiliary engines in this example reduced 10 battery cycles.



Configurations with Batteries

Figure 9. Number of cycles that each battery required in each configuration for a 10% lower demand.

The use of the PEMFC does not present a significant reduction of cycles in configurations with auxiliary engines. In fact, for LFP batteries, the use of the PEMFC increased the number of cycles in configurations including main and auxiliary engines. For LTO batteries, on the other hand, the PEMFC helped to lightly reduce the number of cycles. Beside that, in configurations with only main generators and batteries, the highest reduction was achieved through the use of PEMFC with a hydrogen tank containing 278 kg. This reduction reached at least 3 cycles.

Figure 9 also shows that the size of the hydrogen tank plays no part in reducing the number of cycles. Compared to the case with main and auxiliary generators and batteries, the use of the PEMFC achieved, for LTO-948 kW and LTO-1121 kW batteries, a reduction of 2 and 1 cycles, respectively, for the best case with 565 kg of hydrogen storage. The LTO-948 kW charged and discharged 21 times, whereas the LTO-1121 kW charged and discharged 18 times. For LNMC batteries, the use of the PEMFC did not cause a considerable reduction in terms of cycles. The use of fuel cells with a hydrogen tank containing 565 kg of hydrogen reduced 1 cycle for the LNMC-3119 kW but did not reduce the number of cycles for LNMC-2410 kW battery. The LNMC-2410 kW had 9 cycles during the mission, whereas the LNMC-3119 kW cycled 7 times.

Based on the cycles' limitation shown in Figure 4, the number of missions that each battery can pursue is easily calculated. This number of missions is considered for a battery operating at a temperature of 25 °C and a DoD of 100%. The number of cycles increased with the decrease of the DoD. The DoD used in this work was 60%, so, the number of missions calculated can be higher for a real case. It is important to mention that aging is not considered here, so a real battery would see a reduction in capacity across the battery's cycle life, which would probably lead to an increase in the number of cycles per mission; this effect is not estimated in this paper.

Considering that, when the battery reaches this theoretical maximum number of cycles, the battery should be replaced, we can estimate the number of missions for each battery. The highest number of missions is achieved by LTO batteries. In a PSV with the LTO-948 kW, the number of missions varies from 405, for the configuration with main engines and batteries, to 714, for the best configuration in terms of cycles in which main and auxiliary engines power the ship with the PEMFC and 565 kg of hydrogen. With the LTO-1121 kW, these numbers vary from 468 to 833 for the same configurations. For this battery, this maximum number of missions is also reached for a configuration including 486 kg of hydrogen. In terms of years that the PSV can operate for, the PSV can operate for between 5.2 and 9.2 years with the LTO-948 kW before replacing the batteries, and from 6 to 10.7 years with the LTO-1121 kW. The calculation in years assumes that one mission starts just after the other during the whole year.

LNMC batteries achieved the lowest number of cycles in the same configuration as the LTO batteries. The number of missions for these batteries varied from 187 to 333 for the LNMC-2,410 kW battery. In terms of years, this represents a variation from 2.4 to 4.3 years. For the LNMC-3119 kW battery, the numbers vary from 230 to 428 missions and from 3 to 5.5 years.

LFP batteries lasted for the lowest number of missions. As these batteries have their lowest number of cycles achieved in the configuration with main and auxiliary engines and without fuel cells, the highest number of missions will be mentioned for this configuration. LFP-1284 kW have the number of missions varying from 128 to 300. In terms of years, these numbers vary from 1.65 to 3.9. The LFP-1510 kW would be able to pursue from 144 to 327 missions, depending on which configuration they were installed. In terms of years the numbers would vary from 1.8 to 4.2.

Beside the analysis of emission and battery cycles for each configuration, an analysis in terms of energy generated also can be traced. A comparison of the energy generated by the main generators in all configurations can be made to determine the contribution of these generators to the total energy generated. This is important, because reducing this participation can lead to a broader period between maintenance services.

Considering the 10% lower demand, with only main generators, generators 1, 2, 3, and 4 are responsible for 60.6%, 22.4%, 15.5%, and 1.4% of the demand, respectively. When two auxiliary generators are included, the contribution of the four main generators is reduced by 19.2%. If fuel cells are used instead of auxiliary engines, the reduction of the energy generated by the four generators is lower and varies from 2.8% to 5.6% for 278 kg to 565 kg of stored hydrogen. The configuration with auxiliary generators and fuel cells connected to the main generators reduces the energy generated by the main engines in a variation from 19.4% to 24.6%. As can be seen, this reduction does not simply sum the reduction caused by the insertion of auxiliary generators to the one caused by fuel cells.

In the configurations including only main generators and batteries, there is an increase in the energy generated by the main generators to charge the batteries. Even though the energy generated increases, Figure 8 shows that the configurations including only the main generators and batteries present an emission reduction. It is important to mention that the use of batteries allows for the increase of the generators' minimal point of operation from 10% to 50%. It helps the generators to operate closer to their optimal point, reducing the consumption [66], and, therefore, reducing the emissions even with an increase in energy generation. In terms of energy generated, the increases were 3.24%, 2.89%, 2.44%, 2.43%, 2.09%, and 1.99% for configurations including LFP-1510, LFP-1284, LNMC-2410, LNMC-3110, LTO-1121 and LTO-948, respectively.

For the configuration with main and auxiliary engines and batteries, considerable reductions in the energy generated by the four main generators can be seen for all battery types. The reductions were 19.8%, 20.5%, 21.0%, 21.4%, 30.9%, and 31.4% for LNMC-3119 kW, LTO-1121 kW, LNMC-2410 kW, LTO-948 kW, LFP-1510 kW, and LFP-1284 kW, respectively. The reductions for LFP batteries were the most significant found, with a 10% lower demand.

For configurations including main generators, the PEMFC, and batteries there were minor reductions regarding energy generated. For configurations with 278 kg of stored hydrogen, the reductions were around 0.55% for LNMC and 0.85% for LTO batteries. With LFP batteries, increases of approximately 0.25% were seen. These numbers are much lower when compared to the reduction brought by the same configuration without batteries, in which the reduction reached 2.7%. With an increase in hydrogen storage capacity, reductions around 2% and 3% with batteries and 4.7% and 5.5% for configurations without battery for 486 kg and 565 kg of hydrogen were observed, respectively.

For configurations including main and auxiliary engines, batteries, and the PEMFC, the reductions are substantial; for these configurations, the reductions vary from 18.7%, in the worst case that comprises a LNMC-3119 kW battery and 278 kg of hydrogen, to 25.2%, in the case comprising the same battery with 565 kg of hydrogen.

Finally, it is important to discuss about the contribution of the equipment in each part of the mission. This is shown in Table 6. Since the discussion would be very long if we pointed out the participation of each piece of equipment in each configuration analyzed, Table 6 only shows the data for the configuration including the main and auxiliary generators, LNMC-3119 kW battery, and a 250 kW PEMFC with 565 kg of hydrogen that achieved the highest-emission reduction for a 10% lower demand. The 565 kg of hydrogen finishes in the middle of the DP operation, which is the reason the PEMFC does not appear in the dispatch of Partial Load Voyage or Standby.

	Generators	Equipment Used	Battery
Loading in Port	69.83%	One auxiliary generator PEMFC	30.18%
Laden Voyage	95.52%	Three main generators Two auxiliary generators PEMFC	4.49%
Dynamic Positioning 77.11%		One main generators Two auxiliary generators PEMFC	22.89%
Partial Load Voyage	95.92%	Three main generators Two auxiliary generators	4.08%
Standby	51.60%	Two auxiliary generators	48.40%

Table 6. Energy contribution in percentages for generators and the batteries that compose the configuration that achieved the highest-emission reduction for a 10% lower demand.

6.2. Normal Demand

Figure 10 presents the total emission percentages related to the base case obtained from the configurations operating at normal demand. The results demonstrate that the increase of the hydrogen capacity increases the emission reduction. Moreover, the most significant reductions are achieved with the LNMC-3119 kW in configurations with main and auxiliary generators, PEMFC and batteries with 565 kg of stored hydrogen. This configuration achieved a reduction of 9.32%. With 486 kg and 278 kg of hydrogen, the reductions are 8.64% and 7.08%.

Figure 10 presents the total emissions of configurations without the PEMFC written horizontally. As can be seen, auxiliary engines provided a better reduction of emissions in configurations without batteries and fuel cells. In that case, the reductions were close to those achieved in configurations with main generators and batteries. A comparison between configurations including main generators and LFP batteries and those including main and auxiliary engines shows that auxiliary generators can provide better results in terms of emissions if the normal demand is evaluated. In that case, shipowners must decide between batteries and auxiliary engines as configurations with main and auxiliary engines, and batteries do not provide much better emission reduction.

Contrasting the results in Figure 10 with those presented in Figure 8, it can be observed that the emission reductions obtained for the lowest demand are higher than those obtained for the normal demand. Beside that, for normal demand, the configuration including main and auxiliary generators achieved more emission reductions than configurations including main generators and LFP batteries, which did not occur for the 10% lower demand.

Considering the configurations including main and auxiliary diesel engines, PEMFC, and batteries one may see that the battery that provided the highest reduction is the LNMC-3119 kW. This occurs at all levels of hydrogen amounts. Comparing Figure 8 and Figure 10, it can be seen that there is a change in the battery that provides the second largest emission reduction. For normal demand, the battery is the LTO-1121 kW, whereas for a 10% lower demand it is the LNMC-2410 kW.

Regarding the impact of fuel cells in each configuration, when the same amount of hydrogen is considered, it presents emission reductions very close to each other for all configurations. The only configuration that presented lower reductions was the configuration including only main generators. It also happened when the 10% lower demand was considered. With 565 kg of hydrogen, the reductions vary from 4.4% to 5.13%; for 486 kg, the variation is from 3.83% to 4.44%; for 278 kg, the numbers go from 2.22% to 2.76%. The highest impacts were found in configurations including main and auxiliary generators, fuel cells, and the LFP-1510 kW battery with 278 kg and 486 kg of hydrogen. With 565 kg, the most significant reduction is found in the configuration including main generators, fuel cells, and the LNMC-2410 kW battery.

An evaluation of the battery impact in emissions can also be made based on the results in Figure 10. Batteries presented a more significant impact in configurations without auxiliary engines. For all levels of hydrogen, LNMC-3119 batteries are those which present the largest difference in terms of emissions when configurations including only main generators are analyzed with and without a fuel cell. Even though the reduction is much smaller, for configurations including auxiliary generators, the LNMC-3119 kW battery also results in the highest emission reductions.

From Figure 10, it is clear that, in terms of emissions, the use of auxiliary generators combined with a battery is not indicated for normal demand levels; apart from requiring more capital to buy the auxiliary generator, these engines do not present a considerable increase in emission reduction. Further discussions will highlight the importance of auxiliary generators in reducing the number of battery cycles.



Figure 10. Total emission reduction results obtained from the configurations operating at normal demand.

Figure 11 shows the number of cycles of each battery in all configurations analyzed in this work when the normal demand is considered. As in the previous case with the 10% lower demand, the lowest numbers of cycles of LFP batteries are also achieved in the configuration with main and auxiliary engines and batteries. The number of cycles is exactly the same 12 and 11 cycles for LFP-1284 kW and LFP-1510 kW, respectively. The lowest number of cycles for LTO-948 kW and LNMC-2410 kW batteries was achieved with the configuration including main and auxiliary engines and fuel cells with 278 kg and 486 kg of hydrogen, 20 cycles for LTO-948 kW, 17 cycles for LTO-1121 kW, and 8 cycles for LNMC-2410 kW. With the LNMC-3119 kW battery, the lowest number of 7 cycles is obtained with the configuration with main and auxiliary generators and with the configuration that also included the fuel cell and all amounts of hydrogen.

Different from what happened with the 10% lower demand, the largest amount of hydrogen did not achieve the lowest number of cycles in any configuration with a normal demand. Moreover, with the normal demand, the increase from 485 kg to 565 kg caused an increase in one cycle of LTO-1121 kW and LNMC-2410 kW.

The variation from the 10% lower demand to the normal demand leads to an increase in the number of cycles for some batteries. For the LTO-948 kW batter, the number increases in configuration with the main engines and fuel cells with 278 kg and 565 kg of hydrogen. For the LTO-1121 kW battery, the increase happens when the configuration with main engines and fuel cells with 486 kg and 565 kg of hydrogen. The configuration with main engines and fuel cells with 278 kg and 486 kg of hydrogen presents an increase in the number of cycles for LFP-1510 kW, whereas for configuration including LFP-1284 kW the cycles' increase is seen only in configurations including main generators and fuel cells with 565 kg of hydrogen. LNMC batteries presented a lower or identical number of cycles for all configurations. For the configuration including only main generators, the increase of the demand reduces the number of cycles for LTO-948 kW and both LNMC batteries; for the remaining batteries the number of cycles remains the same.



Figure 11. Number of cycles that each battery required in each configuration for normal demand.

As was completed for the 10% lower demand, the number of missions and the number of years can be calculated based on the number of cycles that each battery can pursue. LTO-948 kW can accomplish a number of missions that vary from 428 to 750; in terms of years, the numbers vary from 5.5 to 9.7. LTO-1121 kW can last from 6 to 11.4 years; in terms of missions, the numbers vary from 238 to 441. LFP-1284 kW would be able to pursue from 128 to 300 missions; in terms of hours, these numbers vary from 1.6 to 3.9 years. LFP-1510 kW can last from 1.8 to 4.2 years and pursue a number of missions that varies from 144 to 327. LNMC-2410 kW can accomplish a number of missions that varies from 187 to 375, in years, the numbers vary from 2.4 to 4.8. The number of missions vary from 250 to 428 and the number of years from 3.2 to 5.5 for the LNMC-3119 kW battery.

A comparison between each configuration analyzed and the base case which comprises only main generators can also be traced in terms of energy generated. Considering the normal demand curve, for the configuration with only main generators, generators 1, 2, 3, and 4 are responsible for 58.2%, 20.3%, 17.3%, and 4.2%, respectively. When two auxiliary generators are included, the contribution of the four main generators is reduced by 17.8%. If fuel cells are used instead of the auxiliary engines, the reduction of the energy generated by the four generators is lower and varies from 2.4% to 4.9% for 278kg to 565 kg. The configuration with auxiliary generators and fuel cells connected to the main generators reduced the energy generated by the main engines in a variation from 17.8% to 22.4%. As can be seen, this reduction does not simply sum the reduction caused by the insertion of auxiliary generators to the one caused by fuel cells.

In the configurations including only main generators and batteries, there are no reductions, because, in those cases, energy is generated by the main engines to charge the batteries. However, the relationship between energy generated and emissions must be discussed. As shown in Figure 10, the configurations including only main generators and batteries provide a reduction of emissions. This happens, because, for the configuration with only main generators, there is energy generated from 10% to 40% of the generator rated power, whereas the insertion of batteries allows for a change of the minimum load of generation from 10% to 50%.

For the configuration with main and auxiliary engines and batteries, considerable reductions in the energy generated by the four main generators can be seen for all battery types. The reductions are 17%, 17.3%, 18.7% and 18.8%, 25.4%, and 26% for LNMC-3119 kW, LNMC-2410 kW, LTO-1121 kW, LTO-948 kW, LFP-1510 kW, and LFP-1284 kW. The reductions for LFP batteries are the highest found for the normal demand. These levels of reductions are not seen in configurations including main generators, fuel cells, and batteries. For configurations with 278 kg of hydrogen, there are minor reductions for LNMC and the LTO-948 kW batteries. For other batteries, minor increases are seen. With 486 kg and 565 kg of hydrogen, reductions around 2% and 3% are found, respectively. For configurations including main and auxiliary engines, batteries, and fuel cells, the reductions are substantial. In these cases, the reductions vary from 17.9%, in the worst case that comprises LNMC-2410 kW battery and 278 kg of hydrogen, to 22.9%, in the case comprising LNMC-3119 kW with 565 kg of hydrogen.

The same discussion pursued at the end of the subsection of the 10% lower demand will be performed here for the normal demand. The data of the generators and battery participation to fed the demand is shown in Table 7. For normal demand, we have the same configuration analyzed for a 10% lower demand, this configuration includes main and auxiliary generators, a LNMC-3119 kW battery, and a 250 kW PEMFC with 565 kg of hydrogen. This configuration achieved the highest emission reduction for the normal demand. The 565 kg of hydrogen also finishes during DP operation and, for normal demand, the PEMFC does not participate in the dispatch of Partial Load Voyage or Standby. Comparing Table 6 to Table 7, we can see that the increase of the demand requires four main generators instead of three for a laden voyage.

	Generators	Equipment Used	Battery
Loading in Port	71.70%	Two auxiliary generators PEMFC	28.30%
Laden Voyage	96.10%	Four main generators Two auxiliary generators PEMFC	3.90%
Dynamic Positioning	80.60%	One main generator Two auxiliary generators PEMFC	19.40%
Partial Load Voyage	94.30%	Three main generators Two auxiliary generators	5.70%
Standby	55.80%	Two auxiliary generators	44.20%

Table 7. Energy contributions in percentages of the generators and the battery that compose the configurations that achieved the highest emission reductions for the normal demand.

6.3. 10% Higher Demand

Figure 12 shows the emission reductions obtained from the different power system configurations operating with a 10% higher demand related to the base case including only four main diesel generators. The most significant emission reduction was approximately 8.33% and was achieved with the configuration including main, LNMC-3119 kW batteries and PEMFC with 565 kg of hydrogen. On the other hand, in configurations including auxiliary generators, the same battery also achieves the highest reduction, which is very similar to the highest emission reduction achieved (around 8.32%).

Figure 12 also shows the emission reduction obtained by configurations without fuel cells. It is interesting to note that LNMC batteries presented better results in configurations without auxiliary engines. For other batteries, the use of auxiliary engines presented a small reduction. The configuration including only a main generator and a LNMC-3119 battery presented a reduction of 4.08%, whereas the configuration including auxiliary engines achieved an emission reduction of 3.77%.

Figure 12 also shows the impact of battery connection in configurations with main generators, as well as main and auxiliary generators. In configurations with only main generators, batteries can provide reductions of ship emissions that vary from 2.65% to 4.08%, depending on which battery is chosen. On the other hand, in configurations including main and auxiliary generators, this reduction achieved by the connection of batteries is not significant. Depending on the battery chemistry, the emission reduction numbers compared to the configuration including main and auxiliary generators vary from 0.12% to 1.06%. Considering that the configuration including main and auxiliary generators achieved an emission decrease of 2.75%, the costs of the auxiliary generators and of each battery should be investigated before a final decision is made.





Figure 12. Total emission reduction results obtained from the configurations operating at 10% higher demand.

Figure 13 shows the number of cycles that each battery pursues for all configurations analyzed in this work for a 10% higher demand. In this case, the lowest number of cycles is achieved for the configurations including main and auxiliary engines for every battery analyzed. It is important to note that the use of fuel cells in the configuration including auxiliary and main engines led to an increase in the number of cycles in all amounts of hydrogen stored. Whereas for LNMC and LTO batteries, the numbers do not differ much, for LFP batteries the insertion of fuel cells cause an increase of 25% in the number of cycles.



Configurations with Batteries

Figure 13. Number of cycles that each battery required in each configuration for a 10% higher demand.

Considering the three demand levels, in a 10% higher demand, the insertion of fuel cells brings about, for the first time, an increase in the number of cycles in configurations without auxiliary generators. Here, the configuration including main engines and a fuel cell with a hydrogen tank of 565 kg increased the number of cycles for all batteries, except for the LNMC-2410 kW and the LFP-1284 kW batteries. With these batteries, the number of cycles was kept the same as presented in the configuration with the main engines.

The variation of the demand level from normal demand to a 10% higher demand implies an increase in the number of cycles for all batteries in configurations with main and auxiliary engines and fuel cells with all sizes of hydrogen tanks. For configurations with main and auxiliary engines, the number of cycles increased for LTO-948 kW batteries, for other batteries this configuration did not change the number of cycles.

The impact that the demand variation can cause on configurations with only main generators and batteries varies from one battery to another. From the 10% lower demand to the 10% higher demand, LTO-948 kW reduces the number of cycles from 37 to 33, LTO-1121 kW reduces it from 32 to 29, LFP-1284 kW from 28 to 26, and LFP-1510 kW from 25 to 22. The reduction in LNMC-2410 kW was from 16 to 15 and in LNMC-3119 kW from 13 to 11.

In terms of missions, when a level of power is required such as a 10% higher demand, the numbers vary from 441 to 714 for LTO-948 kW. For LTO-1121 kW, it varies from 500 to 833. LFP-1284 kW and LFP-1,510 kW have the same maximum number of missions as

presented for a 10% lower demand and normal demand, 300 for LFP-1284 kW and 327 for LFP-1510 kW battery; the lowest number on the other hand is higher than other levels of demand, 138 and 156 for these batteries, respectively. For LNMC-2410 kW, the numbers vary from 200 to 333, and, for, LNMC-3119 kW, the numbers vary from 250 to 428. In terms of years, the minimal and maximum values of years are 5.7 and 9.2 for LTO-948 kW. For LTO-1121 kW, they are 6.4 and 10.7. For LFP-1284 kW, they are 1.8 and 3.9 and 2 and 4.2 for LFP-1510 kW. For LNMC-2410 kW and LNMC-3119 kW, the minimal numbers of years are 2.6 and 3.2, whereas the maximum numbers are 4.3 and 5.5, respectively.

Moving to the electrical analysis, in configurations with only main generators, generators 1, 2, 3, and 4 are responsible for 56.1%, 18.8%, 17.5%, and 7.6% of the energy generated, respectively. The connection of auxiliary generators reduces the amount of energy generated by the main generators to 14.5%. When fuel cells are used instead of auxiliary generators, these reductions vary from 2.2% to 4.5%. For configurations including main and auxiliary generators and fuel cells, the reductions vary from 14.8% to 16.9%.

For configurations including only main generators and batteries, there is an increase in energy generation from the main generators. The increases are 1%, 1.5%, 1.5%, 1.7%, 2.4%, and 2.5% for LNMC-3119 kW, LNMC-2410 kW, LTO-948 kW, LTO-1121 kW, LFP-1284 kW, and LFP-1510 kW, respectively.

For configurations including main and auxiliary generators and batteries, the reductions in energy generation from the main generators are 14.8%, 15.4%, 15.5%, 16.1%, 23.5%, and 23.8% for LNMC-2410 kW, LTO-1121 kW, LNMC-3119 kW, LTO-948 kW, LFP-1284 kW, and LFP-1510 kW, respectively. As mentioned previously, LFP batteries have the largest difference between energy charged and discharged; for this configuration, this energy does not come from main generators but from auxiliary generators. For configurations including LFP batteries, the energy from auxiliary generators is around 70% higher than that of the configuration without a battery.

Using the hydrogen tank with 278 kg in configurations with only main generators and batteries provides almost no change in energy generation from main generators. With 486 kg, the reductions in energy generation from the main generators are 1.5%, 1.6%, 2.2%, 2.5%, 2.5%, and 2.9% for LFP-1510 kW, LFP-1284 kW, LTO-1121 kW, LTO-948 kW, LNMC-2410 kW, and LNMC-3119 kW, respectively. With 565 kg, the reductions for LFP-1284 kW, LFP-1510 kW, LTO-1121 kW, LTO-1121 kW, are 2%, 2.3%, 2.8%, 3% 3.1%, and 3.5%, respectively.

For configurations including main and auxiliary generators, fuel cells, and batteries, the use of 278 kg of hydrogen leads to reductions of 14.3%, 15.4%, 15.5%, 15.8%, 16.4%, and 16.5% for LNMC-2410 kW, LNMC-3119 kW, LFP-1510 kW, LTO-1121 kW, LFP-1284 kW, and LTO-948 kW, respectively. With 486 kg, the numbers are 15.3%, 16.9%, 16.9%, 17.2%, 17.6%, and 17.6% for LNMC-2410 kW, LFP-1510 kW, LTO-1121 kW, LTO-948 kW, LNMC-3119 kW, and LFP-1284 kW, respectively. With 565 kg, the numbers for LNMC-2410 kW, LFP-1510 kW, LTO-1121 kW, and LFP-1284 kW, respectively. With 565 kg, the numbers for LNMC-2410 kW, LFP-1284 kW, and LFP-1284 kW, respectively. With 565 kg, the numbers for LNMC-2410 kW, LFP-1284 kW, LFP-1510 kW, LTO-1121 kW, and LTO-948 kW are 15.8%, 17.2%, 17.2%, 17.5%, 17.6%, and 17.6%, respectively. As can be seen, an increase in the amount of hydrogen stored does not produce a reduction in the energy generated by the main generators for all batteries. Moreover, for LFP batteries, the increments in the energy generated from the auxiliary generators are, on average, 21%, 18%, and 13% for 278 kg, 486 kg, and 565 kg, respectively.

The discussion about energy contributions should also be completed for a 10% higher demand. For a 10% higher demand, the configuration that achieved the highest emission reduction is composed by main generators, a LNMC-3119 kW battery, and a 250 kW PEMFC with 565 kg of hydrogen. The contribution percentages of the generators and battery are shown in Table 8. The 565 kg of hydrogen also finishes during DP operation and, for normal demand PEMFC, does not participate in the dispatch of Partial Load Voyage or Standby. Comparing Table 8 to Tables 6 and 7, we can see that the absence of auxiliary generators led to a higher participation of batteries. Moreover, the increase of the demand requires four main generators instead of three for partial load voyage.

	Generators	Equipment Used	Battery
Loading in Port	54.9%	Two auxiliary generators PEMFC	45.1%
Laden Voyage	92.6%	Four main generators Two auxiliary generators PEMFC	7.4%
Dynamic Positioning	73.2%	One main generator Two auxiliary generators PEMFC	26.8%
Partial Load Voyage	91.0%	Three main generators Two auxiliary generators	9.0%
Standby	20.5%	Two auxiliary generators	79.5%

Table 8. Energy contributions in percentages for generators and the batteries that compose the configurations that achieved the highest emission reductions for a 10% higher demand.

6.4. Emission Results Summary

Figure 14 compiles the results obtained in this work. For each configuration, the highest levels of reduction are achieved for the 10% lower demand. As mentioned before, this demand level can result from a ship-speed reduction or from a load disconnection. This demand may also represents a ship with the same characteristics that attend a platform closer to the port. The lowest reductions were achieved by the configurations without batteries, and the highest reductions were achieved by configurations including a LNMC-3119 kW battery.

Summarizing the results, the five best results for each demand level can be discussed. For the 10% lower demand, four of these five highest emissions were achieved with the same configuration that included main and auxiliary engines, batteries, and fuel cells with 565 kg of hydrogen. The third highest emission reduction of these five was achieved with the same configuration but with 486 kg of hydrogen.

With the normal demand, among the five best results were two configurations including main generators, fuel cells, and batteries, while the other three also had auxiliary generators. With the 10% higher demand, the configurations that achieved the five highest emission results are the same as those presented for the normal demand. However, the ranking order was different. For the normal and the 10% higher demand, the use of 486 kg of hydrogen did not achieve one of the five largest emission reductions.

The five lowest emission reductions for the three levels of demand can also be analyzed. For the 10% lower demand, apart from the configuration including only main generators, there is also one including main generators and the PEMFC with 278 kg of hydrogen, one comprising main and auxiliary generators only, and two including main generators and batteries.

The same analysis can be performed for a normal demand. For this demand level, the configurations that achieved the five lowest emission reductions are the same configurations that were presented for a 10% lower demand. Nonetheless, the ranking order is different. For the 10% higher demand, the ranking order is the same as that obtained for normal demand. The configuration including only main generators and a PEMFC with 486 kg of hydrogen achieved the second lowest reduction for all levels of demand.

It is possible to see that, comparing the lowest and the highest emission reductions for the same configuration, these numbers decrease with the increase of the demand, i.e., from 4.52% to 4.08% for main generators, from 1.64% to 1.04% for main and auxiliary generators. For main and fuel cells the variation goes from 4.66% to 4.12% for 278 kg kg of hydrogen, from 4.76% to 4.30% for 486 kg, and from 4.92% to 4.22% for 565 kg. For configuration with main and auxiliary engines and PEMFC, the difference between the highest and the lowest varies from 1.66% to 1.14% for 278 kg, from 1.75% to 1.24% for 486 kg, and from 1.60% to 1.29% for 565 kg.

		LFP-1284kW						LNMC-3119kW				
No Batt			LFI	⊃-1510k\	N		LTO-948kW LNMC-2410kW					W
	Main Generators		Main + Aux. Generators		Main Generators and Fuel Cell + 565kg of H_2			Main + Aux. Generators and Fuel Cell + 565kg of H ₂				
	-10	Normal	+10	-10	Normal	+10	-10	Normal	+10	-10	Normal	+10
Lowest Reduct.	0.00	0.00	0.00	3.72	3.34	2.73	5.06	4.40	4.11	9.09	8.01	7.03
	3.82	2.98	2.65	4.14	3.46	2 <u>.</u> 87	9.15	7.74	6.99	9.54	8.32	7.39
	4.03	3.27	2.69	4.27	3.48	3.00	9.24	8.22	7.13	9.82	8.36	7.62
	4.24	3.58	3.16	4.66	4.01	3.32	9.81	8.47	7.55	10.02	8.72	7.81
	4.49	3.75	3.18	4.72	4.04	3.41	9.88	8.63	7.67	10.04	8.81	7.87
	4.52	3.96	3.61	4.84	4.04	3.48	9.97	8.88	7.89	10.22	8.90	7.93
Highest Reduct.	4.52	4.31	4.08	5.36	4.59	3.77	9.98	9.01	8.33	10.69	9.33	8.32

Figure 14. Compilation of the emission reductions achieved by configurations analyzed in this paper, except those including 278kg and 486kg of hydrogen. The highest and lowest emission reductions are highlighted, numbers are shown as percentages.

As can be seen, for a normal demand and for a 10% lower demand, the use of auxiliary generators always increases the emission reduction obtained from a configuration without auxiliary generators. For a 10% higher demand, there are configurations in which the use of auxiliary generators will not increase the emission reduction.

7. Conclusions

This work presented a study to assess the CO_2 emission reductions obtained from the connection of different solutions in a platform supply vessel (PSV). The power systems studied consisted of auxiliary diesel generators, three battery chemistries, and a 250 kW proton exchange membrane fuel cell (PEMFC). The CO_2 emissions for each possible configuration, considering these technologies, were estimated by simulations performed using the HOMER software.

The reductions achieved numbers higher than 10%. A configuration with main and auxiliary generators, a fuel cell with 565 kg of hydrogen stored, and the LNMC-3119 kW battery reduced 10.69% of the CO_2 emissions for a 10% lower demand. Considering the same demand level and this configuration without a battery, the emission reduction achieved was 9.09%.

The configuration with LNMC-3119 kW also achieved the highest emission reduction for a normal demand and for a 10% higher demand. For normal demand the reduction was 9.33%, for a 10% higher demand it was 8.32%. For configuration without a battery, the reductions were 8.01% and 7.03% for normal demand and 10% higher demand, respectively.

This paper also presents a discussion about the number of cycles pursued by each battery in each configuration and for each demand level. It can be seen that the increase in the demand level from a 10% lower demand to a 10% higher demand leads to a decrease in the number of cycles in configurations without auxiliary generators. In configurations with auxiliary generators, the variation is minimal with the increase of the demand. For some configurations, there was an increase of one cycle, for others, there was a reduction of one cycle. Beside that, we discussed the importance of auxiliary engines in reducing the number of battery cycles. On average, the use of auxiliary engines reduce in 40% the number of cycles.

From the point of view of emission reduction, the results demonstrate that batteries and auxiliary diesel engines may not present a financially viable solution when used together. The connection of batteries provided a higher emission reduction than the connection of auxiliary engines, but, as mentioned before, the use of only batteries without auxiliary generators has a much higher number of cycles, which will heavily impact costs.

It is important to mention that fuel cells with 565 kg of hydrogen present the largest emission reduction in comparison to batteries and auxiliary engines when these three solutions are analyzed alone. Due to its maturity level, the cost to implement it in a ship is very high.

This paper also discusses the reduction of the participation of main generators in energy dispatches when each solution is connected to the ship's power system. For the three levels of demand, the best solution to reduce the amount of energy generated from main generators is the use of auxiliary generators combined with LFP batteries.

We also discussed the participation of the equipment for the configuration that achieved the highest emission reduction at each level of demand, it can be seen that the configuration without an auxiliary generator relied more on batteries in all parts of the mission. Moreover, it is interesting to note that the absence of two auxiliary generators led to the use of four main generators for partial load voyage.

The main contributions of this paper are the discussion about the influence of each component of battery cycles, the discussion about the influence of each solution on emission reduction, the discussion about the energy reduced in main generators by the connection of each solution, and how each solution behaves depending on the demand level.

Future work should include an economic analysis that would guide the decision related to buying batteries, auxiliary generators, or fuel cells. This economic analysis should also evaluate the fuel reduced by the connection of those solutions. Beside that, future work should also considers the variation of the DoD, i.e., the reduction of the DoD that leads to an increase in the maximum number of cycles. An increase in the power of the PEMFC should also be analyzed, one of the cited papers describes the installation of two 250 kW PEMFC.

Author Contributions: Conceptualization, G.T.T.V.; methodology, G.T.T.V., D.F.P., S.I.T. and K.S.K.; formal analysis, G.T.T.V. and D.F.P.; model development G.T.T.V. and D.F.P., based on discussions with M.B.C.S., J.M.G. and B.S.C.; simulations, G.T.T.V. and D.F.P.; writing–original draft preparation, G.T.T.V. and D.F.P.; writing–review and editing, G.T.T.V., D.F.P., S.I.T., K.S.K., M.B.C.S., J.M.G. and B.S.C.; supervision, M.B.C.S., J.M.G. and B.S.C.; funding acquisition, B.S.C. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Acknowledgments: The authors would like to gratefully acknowledge the support of the RCGI—the Research Centre for Greenhouse Gas Innovation, hosted by the University of São Paulo (USP) and sponsored by the FAPESP—the São Paulo Research Foundation (2014/50279-4 and 2020/15230-5) and Shell Brasil, and the strategic importance of the support given by ANP (Brazil's National Oil, Natural Gas and Biofuels Agency) through the R&D levy regulation. G.T.T.V. acknowledges the financial support of Coordenação de Aperfeiçoamento de Pessoal de Nível Superior—Brazil (CAPES)—Finance Code 001 and Conselho Nacional de Desenvolvimento Científico e Tecnológico (CNPq), process 201674/2020-3. M.B.C.S. thanks the Brazilian National Council for Scientific and Technological Development (CNPq) for financial support in the form of a productivity grant, number 309838/2020-7. B.S.C. thanks the Brazilian National Council for Scientific and Development (CNPq) for financial support in the form of a productivity grant, number 309838/2020-7. B.S.C. thanks the Brazilian National Council for Scientific and Technological Development (CNPq) for financial support in the form of a productivity grant, number 309838/2020-7. B.S.C. thanks the Brazilian National Council for Scientific and Technological Development (CNPq) for financial support in the form of a productivity grant, number 309838/2020-7. B.S.C. thanks the Brazilian National Council for Scientific and Technological Development (CNPq) for financial support in the form of a productivity grant, number 309838/2020-7. B.S.C. thanks the Brazilian National Council for Scientific and Technological Development (CNPq) for financial support in the form of a productivity grant, number 309838/2020-7. B.S.C. thanks the Brazilian National Council for Scientific and Technological Development (CNPq) for financial support in the form of a productivity grant, number 314221/2021-2.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Cushman, J.H., Jr. World Agrees to Cut Shipping Emissions 50 Percent by 2050. 2018. Available online: https://insideclimatenews.org/news/13042018/ocean-shipping-imo-agreement-reduce-climate-change-emissions-fuel-oil-zero-carbon-clean-energy-technology (accessed on 10 November 2020).
- Timperley, J. In-Depth: Will Countries Finally Agree a Climate Deal for Shipping? 2018. Available online: https://www.carbonbrief.org/in-depth-will-countries-finally-agree-climate-deal-for-shipping (accessed on 20 August 2020).
- Monforti-Ferrario, F.; Oreggioni, G.; Schaaf, E. Fossil CO₂ & GHG Emissions of All World Countries; Technical report; European Commission: Brussels, Belgium, 2018.
- European Commission. CO₂ Time Series 1990–2015 Per Region/Country. 2016. Available online: https://edgar.jrc.ec.europa.eu/ overview.php?v=CO2ts1990-2015 (accessed on 20 February 2020).
- 5. Hsieh, C.W.C.; Felby, C. Biofuels for the Marine Shipping Sector; Technical Report; IEA Bioenergy: Paris, France, 2017.
- 6. Miyazaki, M.R.; Sørensen, A.J.; Vartdal, B.J. Reduction of Fuel Consumption on Hybrid Marine Power Plants by Strategic Loading With Energy Storage Devices. *IEEE Power Energy Technol. Syst. J.* 2016, *3*, 207–217. [CrossRef]
- Peralta P, C.O.; Vieira, G.T.; Meunier, S.; Vale, R.J.; Salles, M.B.; Carmo, B.S. Evaluation of the CO₂ emissions reduction potential of Li-ion batteries in ship power systems. *Energies* 2019, *12*, 375. [CrossRef]
- 8. Rafiei, M.; Boudjadar, J.; Khooban, M.H. Energy Management of a Zero-Emission Ferry Boat with a Fuel Cell-Based Hybrid Energy System: Feasibility Assessment. *IEEE Trans. Ind. Electron.* **2020**, *68*, 1739–1748. [CrossRef]
- D'Agostino, F.; Massucco, S.; Schiapparelli, G.P.; Silvestro, F. Modeling and Real-Time Simulation of a DC Shipboard Microgrid. In Proceedings of the 2019 21st European Conference on Power Electronics and Applications (EPE'19 ECCE Europe), Genova, Italy, 3–5 September 2019; pp. 1–6. [CrossRef]
- Simonsen, M.; Gössling, S.; Walnum, H.J. Cruise ship emissions in Norwegian waters: A geographical analysis. *J. Transp. Geogr.* 2019, 78, 87–97. [CrossRef]
- 11. Huang, L.; Wen, Y.; Zhang, Y.; Zhou, C.; Zhang, F.; Yang, T. Dynamic calculation of ship exhaust emissions based on real-time AIS data. *Transp. Res. Part D Transp. Environ.* **2020**, *80*, 102277. [CrossRef]
- 12. Chen, D.; Zhao, Y.; Nelson, P.; Li, Y.; Wang, X.; Zhou, Y.; Lang, J.; Guo, X. Estimating ship emissions based on AIS data for port of Tianjin, China. *Atmos. Environ.* **2016**, *145*, 10–18. [CrossRef]
- 13. Dragović, B.; Tzannatos, E.; Tselentis, V.; Meštrović, R.; Škurić, M. Ship emissions and their externalities in cruise ports. *Transp. Res. Part D Transp. Environ.* **2018**, *61*, 289–300. [CrossRef]
- Nunes, R.; Alvim-Ferraz, M.; Martins, F.; Sousa, S. Assessment of shipping emissions on four ports of Portugal. *Environ. Pollut.* 2017, 231, 1370–1379. [CrossRef]
- 15. Goldsworthy, L.; Goldsworthy, B. Modelling of ship engine exhaust emissions in ports and extensive coastal waters based on terrestrial AIS data—An Australian case study. *Environ. Model. Softw.* **2015**, *63*, 45–60. [CrossRef]
- 16. Jaurola, M.; Hedin, A.; Tikkanen, S.; Huhtala, K. Optimising design and power management in energy-efficient marine vessel power systems: A literature review. *J. Mar. Eng. Technol.* **2018**, *18*, 92–101. [CrossRef]
- Razmjoo, A.; Kaigutha, L.G.; Rad, M.V.; Marzband, M.; Davarpanah, A.; Denai, M. A Technical analysis investigating energy sustainability utilizing reliable renewable energy sources to reduce CO₂ emissions in a high potential area. *Renew. Energy* 2020, 164, 46–57. [CrossRef]
- Moria, H.; Elbreki, A.; Ahmed, A.M.; Elmnifi, M. Optimization and performance evaluation of hybrid renewable system for minimizing CO₂ emissions in Libya: Case study. *Int. J. Renew. Energy Res. (IJRER)* 2020, 10, 1725–1734.
- Gospodinova, D.; Dineff, P.; Milanov, K. Greenhouse Gas Emissions Assessment After Renewable Energy Sources Implementation In Bulgarian Grid-Connected Single-Family Houses By HOMER Pro Software. In Proceedings of the 2020 12th Electrical Engineering Faculty Conference (BulEF), Varna, Bulgaria, 9–12 September 2020; pp. 1–6.
- 20. Batista, A.H.; Ribeiro, E.V.; Naturesa, J.; Salles, M.B. Evaluation of the CO₂ Emissions and Energy Efficiency of the Train Intercidades Campinas-São Paulo with Fuel Cells and Batteries. *An. Soc. Bras. Autom.* **2020**, 2.
- Zahboune, H.; Zouggar, S.; Krajacic, G.; Varbanov, P.S.; Elhafyani, M.; Ziani, E. Optimal hybrid renewable energy design in autonomous system using Modified Electric System Cascade Analysis and Homer software. *Energy Convers. Manag.* 2016, 126, 909–922. [CrossRef]
- 22. Taheri, S.I.; Vieira, G.G.; Salles, M.B.; Avila, S.L. A trip-ahead strategy for optimal energy dispatch in ship power systems. *Electr. Power Syst. Res.* **2021**, *192*, 106917. [CrossRef]
- Vieira, G.; Salles, M.; Monaro, R.M.; Carmo, B.S. CO₂ Emission and Fuel Consumption Evaluation for Variable-Speed Diesel Generators and DC grids for Ship Power Systems. In Proceedings of the 2019 International Conference on Clean Electrical Power (ICCEP), Otranto, Italy, 2–4 July 2019; pp. 544–549.
- 24. Diab, F.; Lan, H.; Ali, S. Novel comparison study between the hybrid renewable energy systems on land and on ship. *Renew. Sustain. Energy Rev.* **2016**, *63*, 452–463. [CrossRef]
- Dawoud, S.M. Techno-Economic and Sensitivity Analysis of Hybrid Electric Sources on Off-Shore Oil Facilities. *Energy* 2021, 227, 120391. [CrossRef]

- Hlal, M.I.; Ramachandaramurthy, V.K.; Sarhan, A.; Pouryekta, A.; Subramaniam, U. Optimum battery depth of discharge for off-grid solar PV/battery system. J. Energy Storage 2019, 26, 100999. [CrossRef]
- Xie, S.; Hu, X.; Zhang, Q.; Lin, X.; Mu, B.; Ji, H. Aging-aware co-optimization of battery size, depth of discharge, and energy management for plug-in hybrid electric vehicles. J. Power Sources 2020, 450, 227638. [CrossRef]
- Morales Vásquez, C.A. A methodology to Select the Electric Propulsion System for Platform Supply Vessels (PSV). Ph.D. Thesis, Universidade de São Paulo, São Paulo, Brazil, 2014.
- Vieira, G.; Peralta, C.; Salles, M.; Carmo, B. Reduction of CO₂ emissions in ships with advanced Energy Storage Systems. In Proceedings of the 2017 6th International Conference on Clean Electrical Power (ICCEP), Santa Margherita Ligure, Italy, 27–29 June 2017; pp. 564–571.
- Vale, R.; Peralta, C.; Vieira, G.; Salles, M.; Carmo, B. Assessment of CO₂ Emissions on Platform Supply Vessels for Distinct Battery Dispatch. In Proceedings of the 2018 7th International Conference on Renewable Energy Research and Applications (ICRERA), Paris, France, 14–17 October 2018; pp. 1287–1291.
- Caterpillar. Datasheet Diesel Generator Caterpillar C280. 2020. Available online: https://pdf.nauticexpo.com/pdf/caterpillarmarine-power-systems/cat-c280-6-genset-spec-sheet/19997-52905.html (accessed on 10 August 2020).
- Caterpillar. Datasheet Diesel Generator Caterpillar C18. 2020. Available online: https://www.pon-cat.com/application/files/ 1615/3555/2399/CM20180613-08777-64587.pdf (accessed on 10 August 2020).
- 33. Fujian Shipbuilding. 87M Platform Supply Vessel. Available online: http://www.fujianshipbuilding.com/87m-platform-supply-vessel (accessed on 26 September 2020).
- Garg, K.; Weingarth, L.; Shah, S. Dynamic positioning power plant system reliability and design. In Proceedings of the Petroleum and Chemical Industry Conference Europe Conference Proceedings (PCIC EUROPE), Rome, Italy, 7–9 June 2011; pp. 1–10.
- 35. Tufte, E.D. Impacts of Low Load Operation of Modern Four-Stroke Diesel Engines in Generator Configuration. Master's Thesis, The Norwegian University of Science and Technology, Trondheim, Norway, 2014.
- Koniak, M.; Czerepicki, A. Selection of the battery pack parameters for an electric vehicle based on performance requirements. In Proceedings of the IOP Conference Series, Materials Science and Engineering, Bangkok, Thailand, 21–23 April 2017; pp. 21–23.
- Ralon, P.; Taylor, M.; Ilas, A.; Diaz-Bone, H.; Kairies, K. *Electricity Storage and Renewables: Costs and Markets to 2030*; International Renewable Energy Agency: Abu Dhabi, United Arab Emirates, 2017.
- 38. Buchmann, I. Types of Lithium-ion. 2019. Available online: https://batteryuniversity.com/learn/article/types_of_lithium_ion (accessed on 4 May 2020).
- 39. Farzin, H.; Fotuhi-Firuzabad, M.; Moeini-Aghtaie, M. A practical scheme to involve degradation cost of lithium-ion batteries in vehicle-to-grid applications. *IEEE Trans. Sustain. Energy* **2016**, *7*, 1730–1738. [CrossRef]
- 40. Xu, B.; Oudalov, A.; Ulbig, A.; Andersson, G.; Kirschen, D.S. Modeling of lithium-ion battery degradation for cell life assessment. *IEEE Trans. Smart Grid* 2016, *9*, 1131–1140. [CrossRef]
- 41. Henze, V. Battery Power's Latest Plunge in Costs Threatens Coal, Gas. 2019. Available online: https://about.bnef.com/blog/ battery-powers-latest-plunge-costs-threatens-coal-gas/ (accessed on 20 August 2020).
- Henze, V. Battery Pack Prices Fall as Market Ramps up with Market Average at \$156/kWh in 2019. 2019. Available online: https: //about.bnef.com/blog/battery-pack-prices-fall-as-market-ramps-up-with-market-average-at-156-kwh-in-2019/ (accessed on 20 August 2020).
- 43. Xiong, R.; Li, L.; Tian, J. Towards a smarter battery management system: A critical review on battery state of health monitoring methods. *J. Power Sources* 2018, 405, 18–29. [CrossRef]
- 44. Vogler, F.; Sattler, G. Hydrogen-fueled marine transportation. In *Compendium of Hydrogen Energy*, 1st ed.; Ball, M., Basile, A., Veziroglu, T.N., Eds.; Woodhead Publishing: Oxford, UK, 2016; Chapter 3.
- 45. Inal, O.B.; Deniz, C. Assessment of fuel cell types for ships: Based on multi-criteria decision analysis. *J. Clean. Prod.* 2020, 265, 121734. [CrossRef]
- 46. De Troya, J.J.; Álvarez, C.; Fernández-Garrido, C.; Carral, L. Analysing the possibilities of using fuel cells in ships. *Int. J. Hydrogen Energy* **2016**, *41*, 2853–2866. [CrossRef]
- 47. Sapre, S.; Pareek, K.; Vyas, M. Investigation of structural stability of type IV compressed hydrogen storage tank during refueling of fuel cell vehicle. *Energy Storage* 2020, 2, e150. [CrossRef]
- Bannenberg, L.; Heere, M.; Benzidi, H.; Montero, J.; Dematteis, E.; Suwarno, S.; Jaroń, T.; Winny, M.; Orłowski, P.; Wegner, W.; et al. Metal (boro-) hydrides for high energy density storage and relevant emerging technologies. *Int. J. Hydrogen Energy* 2020, 45, 33687–33730. [CrossRef]
- 49. Tarasov, B.P.; Fursikov, P.V.; Volodin, A.A.; Bocharnikov, M.S.; Shimkus, Y.Y.; Kashin, A.M.; Yartys, V.A.; Chidziva, S.; Pasupathi, S.; Lototskyy, M.V. Metal hydrogen storage and compression systems for energy storage technologies. *Int. J. Hydrogen Energy* **2020**, *46*, 13647–13657. [CrossRef]
- 50. de Melo Furtado, J.G.; Gatti, G.C.; Serra, E.T.; de Almeida, S.C.A. Performance analysis of a 5 kW PEMFC with a natural gas reformer. *Int. J. Hydrogen Energy* **2010**, *35*, 9990–9995. [CrossRef]
- 51. Yartys, V.A.; Lototsky, M.V. An Overview of Hydrogen Storage Methods. In *Hydrogen Materials Science and Chemistry of Carbon Nanomaterials;* Springer: Dordrecht, The Netherlands, 2005; pp. 75–104.
- 52. Larminie, J.; Dicks, A. Fuel Cells Systems Explained, 2nd ed.; John Wiley & Sons: Chichester, UK, 2003.

- Sandia. Development of a Containerized 100 kW Fuel Cell System for Maritime Applications. 2019. Available online: https://www.energy.gov/sites/prod/files/2019/10/f68/fcto-h2-at-ports-workshop-2019-viii1-klebanoff.pdf (accessed on 10 December 2020).
- 54. FuelCellWorks. TECO 2030 Launches Fuel Cell Container. 2021. Available online: https://fuelcellsworks.com/news/teco-2030-launches-fuel-cell-container/ (accessed on 10 April 2021).
- 55. Díaz-de Baldasano, M.C.; Mateos, F.J.; Núñez-Rivas, L.R.; Leo, T.J. Conceptual design of offshore platform supply vessel based on hybrid diesel generator-fuel cell power plant. *Appl. Energy* **2014**, *116*, 91–100. [CrossRef]
- Brooker, A.; Haraldsson, K.; Hendricks, T.; Johnson, V.; Kelly, K.; Kramer, B.; Markel, T.; O'Keefe, M.; Sprik, S.; Wipke, K.; et al. ADVISOR Advanced Vehicle Simulator. 2013. Available online: http://adv-vehicle-sim.sourceforge.net/ (accessed on 10 December 2020).
- 57. NREL. Vehicle Technology Simulation and Analysis Tools. Available online: https://www.nrel.gov/transportation/systemsanalysis-tools.html (accessed on 10 December 2020).
- LabTech Int. Co. Ltd. HBond—7000L. 2005. Available online: http://www.labtech-hydrogen.com/index.php?page=HS3 (accessed on 29 August 2020).
- 59. Giap, V.T.; Lee, Y.D.; Kim, Y.S.; Ahn, K.Y. A novel electrical energy storage system based on a reversible solid oxide fuel cell coupled with metal hydrides and waste steam. *Appl. Energy* **2020**, *262*, 114522. [CrossRef]
- Hexagon Purus. Hydrogen Storage and Transportation Systems—Hydrogen Type 4 Cylinders. 2020. Available online: https://www.hexagonxperion.com/fileadmin/user_upload/xperion_Energy__Environment/Datenblaetter_-_xperion_EE/ Hexagon_xperion_Datenblaetter/Hexagon-Purus-Hydrogen-storage-and-transformation-systems.pdf (accessed on 26 October 2020).
- 61. Hexagon Purus. X-Store[®] Gas Container Modules, Version ADR V2. 2020. Available online: https://www.hexagonxperion.com/ fileadmin/user_upload/xperion_Energy___Environment/Hexagon_Datenblaetter/Hexagon-Purus-X-STORE-Gas_Container-Modules-Version-ADR-V2-Full-Carbon-Design-500Bar-H2.pdf (accessed on 26 October 2020).
- Commercial Battery. LTO 1865 Rechargeable Cell: 2.4V 1300 mAh. Available online: https://www.batteryspace.com/LTO-1865-Rechargeable-Cell-2.4V-1300-mAh-39A-rated-3.12Wh.aspx (accessed on 4 February 2021).
- 63. Commercial Battery. 18500 3.2 Volt LiFePO4 Battery (800 mAh). Available online: https://www.amazon.com/18500-3-2-Volt-LiFePO4-Battery/dp/B01L7V2DAA (accessed on 4 February 2021).
- 64. Commercial Battery. LiNiMnCo Rechargeable 22650 Cell: 3.7V 2500 mAh. Available online: https://www.batteryspace.com/ linimnco-rechargeable-22650-cell-3-7v-2500-mah-9wh-3-75a-rated---un38-3-passed.aspx (accessed on 4 February 2021).
- 65. Commercial Battery. LiNiMnCo 26650 Rechargeable Cell: 3.6V 5Ah. Available online: https://www.batteryspace.com/linimnco-26650-rechargeable-cell-3-6v-5ah-18wh-15a-rated---un38-3-passed--.aspx (accessed on 4 February 2021).
- Alnes, O.; Eriksen, S.; Vartdal, B.J. Battery-Powered Ships: A Class Society Perspective. IEEE Electrif. Mag. 2017, 5, 10–21. [CrossRef]